\documentclass{article}
\usepackage[affil-it]{authblk}
\usepackage{graphicx}
\usepackage[space]{grffile}
\usepackage{latexsym}
\usepackage{textcomp}
\usepackage{longtable}
\usepackage{tabulary}
\usepackage{booktabs,array,multirow}
\usepackage{amsfonts,amsmath,amssymb}
\providecommand\citet{\cite}
\providecommand\citep{\cite}
\providecommand\citealt{\cite}
\usepackage{url}
\usepackage{hyperref}
\hypersetup{colorlinks=false,pdfborder={0 0 0}}
\usepackage{etoolbox}
\makeatletter
\patchcmd\@combinedblfloats{\box\@outputbox}{\unvbox\@outputbox}{}{%
  \errmessage{\noexpand\@combinedblfloats could not be patched}%
}%
\makeatother
% You can conditionalize code for latexml or normal latex using this.
\newif\iflatexml\latexmlfalse
\AtBeginDocument{\DeclareGraphicsExtensions{.pdf,.PDF,.eps,.EPS,.png,.PNG,.tif,.TIF,.jpg,.JPG,.jpeg,.JPEG}}

\usepackage[utf8]{inputenc}
\usepackage[english]{babel}




\begin{document}

\title{Citations}


\author{Jeff Montgomery}
\affil{Affiliation not available}



\date{\today}

\maketitle




\textbf{What's on your mind?}\\
This is an experimental "article".\\
I'll periodically drop citations to papers and news I'm reading, I've read, or I'm still thinking about.
Dropped citations might have comments, links to related articles/writings, or be effectively untouched.

At any point, anyone can comment on anything (please do; it's encouraged).\\
Maybe you have some insights on a paper, a particular concept, or technique.  Maybe you disagree with something.  Maybe you have more recommended reading.  Maybe you just want to say something is "cool".\\
Any kind of comment is good because it adds information to the thought-process or discussion.

As this is a work very much in progress, organizational changes will be made I'm sure, and if anyone has suggestions or wants to build off anything here, go for it.  The goal, while not yet entirely clear, is some kind of radical or innovative collaboration.

\textbf{Physics}

BCS, 1957 (1st microscopic treatment of superconductivity) \cite{Bardeen_1957}

Time Emerges from Quantum Entanglement \cite{Moreva_2014}\\
\href{https://medium.com/the-physics-arxiv-blog/quantum-experiment-shows-how-time-emerges-from-entanglement-d5d3dc850933}{ArXiv blog post}

\textbf{Bibliometrics, Science Policy, Productivity, and Reproducibility}

Metascience to improve and verify reproducibility. \cite{Schooler_2014}

Google Scholar, "Rise of the Rest": \\
shift in citation distributions (between "high impact" and "non-elite" journals)\\
\href{http://arxiv.org/pdf/1410.2217.pdf}{Paper} \\
\href{https://www.authorea.com/users/9932/articles/10529/_show_article}{Authorea Coverage}

Transparency in Science necessary to build public trust:
\href{http://www.nature.com/news/openness-in-science-is-key-to-keeping-public-trust-1.16371?WT.ec_id=NATURE-20141120}{Nature News}


Google Scholar, "On the Shoulders of Giants": \\
increasing citation rates on older papers \\
\href{http://arxiv.org/pdf/1411.0275.pdf}{Paper} \\
\href{http://googlescholar.blogspot.com/2014/11/on-shoulders-of-giants-growing-impact.html}{Google Scholar Blog}

"Superstar" Researchers \cite{Shockley_1957}

Redesign ochem lab course for professional development/independent research \cite{Alaimo_2014}

\textbf{Evolution}

Evolutionary Cell Biology \cite{Lynch_2014}

Nature News on Ancient Hominids' Genomes  \cite{Callaway_2013}


\textbf{Microbiome/virome}

Microbiome regulates permeability of blood-brain barrier in mice \cite{Braniste_2014}

\href{http://www.nature.com/nature/journal/vaop/ncurrent/full/nature13938.html?WT.ec_id=NATURE-20141120}{Viruses may be beneficial in certain crcumstances in mammalian gut}

\selectlanguage{english}
\FloatBarrier
\bibliographystyle{plain}
\bibliography{bibliography/converted_to_latex.bib%
}

\end{CJK}\end{document}

